Author's response to reviews

Title: Spontaneous tibiotalar arthrodesis: a complication of acute tibial osteomyelitis due to Panton Valentine Leukocidin producing Staphylococcus aureus

Authors:  
Dimitri D Ceroni (dimitri.ceroni@hcuge.ch)  
Rebecca R Anderson della Llana (andersondelallana@hcuge.ch)  
Tristan T Zand (tristan.zand@hcuge.ch)  
Léopold L Lamah (leopold.lamah@hcuge.ch)  
Denis D Dominguez (denis.dominguez@hcuge.ch)  
Geraldo G De Coulon (geraldo.decoulon@hcuge.ch)  
Victor V Dubois Ferrière (victor.duboisferriere@hcuge.ch)  
Bernardo B Vargas (bernardo.vergas@hcuge.ch)

Version: 2 Date: 28 February 2012

Author's response to reviews:

Dear Ms Pangan,

Re: Manuscript "Spontaneous tibiotalar arthrodesis: a complication of acute tibial osteomyelitis due to Panton Valentine Leukocidin producing Staphylococcus aureus" by Ceroni D et al

We should like to thank you for e-mail of the 27 february in which you have kindly invited us to submit a new revised version of our above-mentioned manuscript.

As requested, we realized a point-by-point response to all your comments.

We thank you in advance for your consideration of our manuscript and we look forward to the final decision of the editorial committee.

Yours sincerely,